Author's response to reviews

Title: Triglyceride associated polymorphisms of the APOA5 gene have very different allele frequencies in Pune, India compared to Europeans

Authors:

Giriraj R Chandak (chandakgrc@ccmb.res.in)
Kirsten J Ward (kjward@pms.ac.uk)
Chittaranjan S Yajnik (csyajnik@hotmail.com)
Anand N Pandit (kemhrc@pn2 vsnl.net.in)
Ashish Bavdekar (bavdekar@vsnl.com)
Charu V Joglekar (joglekar@pn2 vsnl.net.in)
Caroline H D Fall (chdf@mrc.soton.ac.uk)
Mohan KRISHNA P (mohanp@ccmb.res.in)
Terence J Wilkin (terence.wilkin@phnt.swest.nhs.uk)
Bradley S Metcalf (brad.metcalf@pms.ac.uk)
Michael N Weedon (michael.weedon@pms.ac.uk)
Timothy M Frayling (tim.frayling@pms.ac.uk)
Andrew T Hattersley (andrew.hattersley@pms.ac.uk)

Version: 7 Date: 10 October 2006

Author's response to reviews: see over
For the attention of:-

Dr. Le
Biomed Central Medical Genetics

Please reply to:

Dr. Kristen Ward
Peninsula Medical School
Royal Devon & Exeter Hospital (Wonford)
Barrack Road, Exeter EX2 5DW
Direct Line: +44 (0)1392 406776
Email: kjward@pms.ac.uk

10th October 2006

Regarding: Submission of revised accepted manuscript after formatting changes have been done

Dear Dr. Le,

Thank you for accepting our manuscript and for sending the formatting changes that were required before publication of the paper. We have made the following changes as requested:

1) Dr Krishna’s name is the same in both the manuscript and on the submission system.
2) The running title has been removed from the title page
3) The keywords section has been removed from the title page
4) We have removed all highlighting and tracking from the manuscript
5) The table title is situated above the table and the legend is situated below the table

If there are any problems, please do not hesitate to contact me for them to be rectified

Yours sincerely

Dr. Kirsten Ward